Author's response to reviews

Title: A case-control study of occupational magnetic field exposure and Alzheimer's disease: Results from the California Alzheimer's Disease and Treatment Centers

Authors:

Zoreh Davanipour (zdavani@aol.com)
Chiu-Chen Tseng (annchiuchen@yahoo.com)
Pey-Jiuan Lee (peylee@usc.edu)
Eugene Sobel (sobel55@earthlink.net)

Version: 5 Date: 31 May 2007

Author’s response to reviews:

The revisions requested by the editors in their 05/14/2007 e-mail have been made:

1. the unnecessary capital letters have been removed;
2. the header has been removed;
3. "List of Abbreviations" has been changed to "Abbreviations";
4. the competing interests section has been rewritten as requested;
5. all table titles appear above and all legends or keys appear below the tables;
6. only the contact details for ZD remain and the location has been changed.

Thank you.